Author's response to reviews

Title: Uncovering emotion work in clinical trials: a qualitative study of the experiences and views of staff involved in a type 1 diabetes trial

Authors:

Julia Lawton (J.Lawton@ed.ac.uk)
Jackie Kirkham (J.Kirkham@ed.ac.uk)
David A White (D.A.White@sheffield.ac.uk)
David Rankin (a.d.rankin@ed.ac.uk)
Cooper L Cindy (C.L.Cooper@sheffield.ac.uk)
Heller Simon (S.heller@sheffield.ac.uk)

Version: 2 Date: 15 July 2014

Author's response to reviews:

We no longer write the paper on behalf of the REPOSE group and no reference is made to the name of the trial in the abstract. For this reason, and because the paper is not reporting the findings from the trial but, rather, findings from qualitative interviews with trial staff, we think it would cause confusion to include the trial registration number at the end of the abstract. We have redrafted the acknowledgements section and we have made included the trial registration number there.

We have renamed findings as 'results'

We have added a list of abbreviations

We now mention each author individually in the authors' contributions section